# ECONSTOR

Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Resende, Guilherme Mendes; Ywata de Carvalho, Alexandre; Sakowski, Patrícia Alessandra Morita

# Working Paper Evaluating multiple spatial dimensionsof economic growth in Brazil using spatial panel data models (1970-2000)

Discussion Paper, No. 193

**Provided in Cooperation with:** Institute of Applied Economic Research (ipea), Brasília

*Suggested Citation:* Resende, Guilherme Mendes; Ywata de Carvalho, Alexandre; Sakowski, Patrícia Alessandra Morita (2015) : Evaluating multiple spatial dimensionsof economic growth in Brazil using spatial panel data models (1970-2000), Discussion Paper, No. 193, Institute for Applied Economic Research (ipea), Brasília

This Version is available at: https://hdl.handle.net/10419/220281

### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



Originally published by Ipea in May 2013 as number 1830a of the series Texto para Discussão.



EVALUATING MULTIPLE SPATIAL DIMENSIONS OF ECONOMIC GROWTH IN BRAZIL USING SPATIAL PANEL DATA MODELS (1970-2000)

Guilherme Mendes Resende Alexandre Xavier Ywata de Carvalho Patrícia Alessandra Morita Sakowski





Originally published by Ipea in May 2013 as number 1830a of the series Texto para Discussão.

Brasília, January 2015

EVALUATING MULTIPLE SPATIAL DIMENSIONSOF ECONOMIC GROWTH IN BRAZIL USING SPATIAL PANEL DATA MODELS (1970-2000)

Guilherme Mendes Resende<sup>1</sup> Alexandre Xavier Ywata de Carvalho<sup>2</sup> Patrícia Alessandra Morita Sakowski<sup>3</sup>

<sup>1.</sup> Institute for Applied Economic Research (Ipea)/Government of Brazil.

<sup>2.</sup> Institute for Applied Economic Research (Ipea)/Government of Brazil.

<sup>3.</sup> Institute for Applied Economic Research (Ipea)/Government of Brazil.

### Federal Government of Brazil

Secretariat of Strategic Affairs of the Presidency of the Republic Minister Roberto Mangabeira Unger



A public foundation affiliated to the Secretariat of Strategic Affairs of the Presidency of the Republic, Ipea provides technical and institutional support to government actions – enabling the formulation of numerous public policies and programs for Brazilian development – and makes research and studies conducted by its staff available to society.

President

Sergei Suarez Dillon Soares

**Director of Institutional Development** Luiz Cezar Loureiro de Azeredo

Director of Studies and Policies of the State, Institutions and Democracy Daniel Ricardo de Castro Cerqueira

Director of Macroeconomic Studies and Policies Cláudio Hamilton Matos dos Santos

Director of Regional, Urban and Environmental Studies and Policies Rogério Boueri Miranda

**Director of Sectoral Studies and Policies, Innovation, Regulation and Infrastructure** Fernanda De Negri

**Director of Social Studies and Policies, Deputy** Carlos Henrique Leite Corseuil

Director of International Studies, Political and Economic Relations Renato Coelho Baumann das Neves

**Chief of Staff** Ruy Silva Pessoa

Chief Press and Communications Officer João Cláudio Garcia Rodrigues Lima

URL: http://www.ipea.gov.br Ombudsman: http://www.ipea.gov.br/ouvidoria

### **DISCUSSION PAPER**

A publication to disseminate the findings of research directly or indirectly conducted by the Institute for Applied Economic Research (Ipea). Due to their relevance, they provide information to specialists and encourage contributions.

© Institute for Applied Economic Research - ipea 2015

Discussion paper / Institute for Applied Economic

Research.- Brasília : Rio de Janeiro : Ipea, 1990-

ISSN 1415-4765

1. Brazil. 2. Economic Aspects. 3. Social Aspects. I. Institute for Applied Economic Research.

CDD 330.908

The authors are exclusively and entirely responsible for the opinions expressed in this volume. These do not necessarily reflect the views of the Institute for Applied Economic Research or of the Secretariat of Strategic Affairs of the Presidency of the Republic.

Reproduction of this text and the data it contains is allowed as long as the source is cited. Reproductions for commercial purposes are prohibited.

JEL: C23, O18, R11.

# CONTENTS

ABSTRACT
SINOPSE
1 MOTIVATION7
2 THE SPATIAL SCOPE OF ECONOMIC GROWTH DETERMINANTS: POTENTIAL THEORETICAL REASONS FOR DIFFERENT RESULTS ACROSS MODELS ESTIMATED AT DIFFERENT SPATIAL SCALES
3 SPATIAL PANEL DATA MODELS12
4 DATA
5 EMPIRICAL RESULTS19
6 DISCUSSION
7 CONCLUSIONS
REFERENCES
APPENDIX

# ABSTRACT

The goal of this paper is to evaluate the results of regional economic growth estimates at multiple spatial scales using spatial panel data models. The spatial scales examined are minimum comparable areas, micro-regions, meso-regions and states over the period between 1970 and 2000. Alternative spatial panel data models with fixed effects were systematically estimated across those spatial scales to demonstrate that the estimated coefficients change with the scale level. The results show that the conclusions obtained from growth regressions are dependent on the choice of spatial scale. First, club convergence hypothesis cannot be rejected suggesting there are differences in the convergence processes between the north and south in Brazil. Moreover, the positive average-years-of-schooling coefficient gets larger as more aggregate spatial scales are used. Transportation costs effect is positive and statistically significant to economic growth only at the state level. Population density coefficients show that higher populated areas are harmful to economic growth demonstrating somehow that congestion effects are operating at the MCA, micro-regional and meso-regional spatial scales, but their magnitudes vary across the geographic scales. Finally, the values of spatial spillovers coefficients also vary according to the spatial scale under analysis. In general, such coefficients are statistically significant at the MCA, micro-regional and meso-regional levels; but, at state level those coefficients are no longer statistically significant suggesting that spatial spillovers are bounded in space.

Keywords: spatial externalities; economic growth; spatial scales; spatial panel; Brazil.

### SINOPSE

O objetivo deste estudo consiste em avaliar os resultados de estimações de crescimento econômico regional em múltiplas escalas espaciais utilizando modelos de painel espacial. As escalas espaciais examinadas são áreas mínimas comparáveis, microrregiões, mesorregiões e estados no período entre 1970 e 2000. Modelos alternativos de painel espacial com efeitos fixos foram estimados sistematicamente nessas escalas espaciais para demonstrar que os coeficientes estimados variam de acordo com a escala utilizada. Os resultados mostram que as conclusões obtidas a partir de regressões de cre scimento dependem da escolha da escala espacial. Primeiramente, a hipótese de convergência de clube não pode ser rejeitada, sugerindo haver diferenças nos processos de convergência entre o norte e sul do Brasil. Além disso, o coeficiente positivo da média de anos de escolaridade aumenta quanto mais agregada a escala espacial utilizada. O efeito de custos de transporte

é positivo e estatisticamente significante para o crescimento econômico apenas no nível do estado. Os coeficientes da densidade populacional mostram que áreas mais densamente povoadas são prejudiciais para o crescimento econômico, sugerindo efeitos de congestionamento no nível de AMC, microrregiões e mesorregiões, mas a magnitude desses coeficientes varia de acordo com a escala geográfica. Finalmente os coeficientes de transbordamento espacial também variam conforme a escala espacial sob análise. Em geral, esses coeficientes são estatisticamente significativos nos níveis de AMC, microrregião e mesorregião; mas, no nível estadual, deixam de ser estatisticamente significativos, sugerindo que transbordamentos espaciais são limitados no espaço.

**Palavras-chave:** externalidades espaciais; crescimento econômico; escalas espaciais; painel espacial; Brasil.

# **1 MOTIVATION**

During the last two decades, an increasing dissemination of spatial econometrics techniques has been observed among regional scientists, economists and researchers in several fields (Anselin, 1988; LeSage, 1999; Conley, 1999). The vast research of applied spatial econometrics on the interdependencies among spatial units and their effects on, among others, regional economic growth, trade flows, knowledge spillovers, migration, housing prices, tax interactions, city's growth controls (e.g., López-Bazo *et al.*, 2004; Gamboa, 2010; Fischer *et al.*, 2009; LeSage and Pace, 2008, Jeanty *et al.*, 2010; Gérard *et al.*, 2010; Brueckner, 1998) is well known. However, this literature still lacks a better understanding of the potential reasons why models estimated at different geographic scales yield different results in the context of regional economic growth empirics.

Of note, Resende (2011) investigates the determinants of Brazilian regional economic growth at a variety of geographic scales using a cross-sectional data set over the 1990s period. Moreover, Resende (2013) advances the growth literature by using panel data models across several spatial scales; but the process of economic growth in Brazil is only examined using non-spatial panel data models. This investigation refers back to the Modifiable Areal Unit Problem (MAUP),<sup>1</sup> but it sheds new light on a core problem in the literature related to regional economic growth. The choice of the spatial scale of analysis is a problematic issue in applied research (Behrens and Thisse, 2007). In this sense, the paper seeks to investigate to what extent ambiguities about spatial scale undermine, or inform, our understanding of regional growth determinants and convergence.<sup>2</sup>

Except from Resende (2011) and Resende (2013)<sup>3</sup> the studies thus far have only investigated the determinants of economic growth at a single spatial scale to infer the consistency of spatial growth models with reality (e.g., Rey and Montouri, 1999; Fingleton, 1999; López-Bazo *et al.*, 2004; Ertur and Koch, 2007; Elhorst *et al.*, 2010, Fischer, 2011). For instance, Elhorst *et al.* (2010) employ spatial econometric techniques to focus on time-space models; but they only examine the process of economic growth at one single

<sup>1.</sup> MAUP is associated to the uncertainties on the choice of alternative number of zones (or zoning systems) and the implications that this holds for spatial analysis (Openshaw and Taylor, 1981).

<sup>2.</sup> It is worth noting that there has been a growing empirical literature on the analysis of MAUP in several areas of urban and regional economics such as Yamamoto (2008), Briant *et al.* (2010), Fingleton (2011) and Menon (2012).

<sup>3.</sup> Of note, Ávila and Monastério (2008) analyse MAUP on per capita income convergence process in the Rio Grande do Sul state in Brazil using two geographic scales (municipalities and "Conselhos Regionais de Desenvolvimento"/COREDEs).

spatial scale. Thus, the goal of this paper is to evaluate the results of regional economic growth estimates at multiple spatial scales using alternative spatio-temporal models recently proposed in the spatial econometrics literature.

The spatial scales examined are minimum comparable areas (referred to as municipalities), micro-regions, meso-regions and states, which are often employed in the empirical literature about Brazil and cover the period between 1970 and 2000. To our knowledge, this is the first study of regional economic growth exploring both time and different spatial scale dimensions in the context of spatial panel data models. Previous studies only investigate both time and space at a single spatial scale. The idea of this paper is to systematically repeat a spatial panel data model originally developed to examine this phenomenon at a single geographic scale across multiple scales. Initially, this approach lead us to the investigation of the measurement issue that might cause variability in regional economic growth estimates due to the use of different spatial scales, likely due to the MAUP. However, it is important to bear in mind that it might be the case that structural (theory based) issues may be underlying economic growth at different scales and, thus, we provide some theoretical arguments for such variability in empirical results found across different geographic scales.<sup>4</sup>

The paper is organized as follows. Section 2 provides a discussion on the potential theoretical reasons for different results found across economic growth models estimated at different spatial scales. Section 3 describes the spatial panel data models employed in the empirical analysis. In section 4, the data set and the spatial scales are shown. Section 5 presents the results and the respective discussion follows in Section 6. Section 7 concludes the paper.

### 2 THE SPATIAL SCOPE OF ECONOMIC GROWTH DETERMINANTS: POTENTIAL THEORETICAL REASONS FOR DIFFERENT RESULTS ACROSS MODELS ESTIMATED AT DIFFERENT SPATIAL SCALES

In the mainstream of economics, economic growth theories provide several factors that may have been responsible for driving regional performance. The debate on long run economic growth determinants came with Solow's (1956) growth model and has been

<sup>4.</sup> In Resende (2011), there is an initial discussion on this issue. Herein, we provide more theoretical arguments for such differences across scales (see Section 2).

augmented by many others by the inclusion of education capital (Mankiw *et al.*, 1992), migration (Barro & Sala-i-Martin, 2003), and growth externalities (López-Bazo *et al.*, 2004; Ertur & Koch, 2007) to cite some. Moreover, the so-called endogenous growth models, pioneered by Romer (1986) and Lucas (1988), seek to explain why differences in per capita income arise and persist over time. Herein, we provide some theoretical reasons to explain how the explanatory variables used in the econometric specifications discussed in the next section may impact economic growth at different spatial scales.

- 1) Physical capital (and the convergence hypotheses). Information on physical capital is often unavailable at subnational levels, and thus this variable is excluded from the set of explanatory variables of regional growth regressions. The lack of availability of physical capital measures at finer regional level is not restricted to Brazil, as noted in Lesage and Fisher (2008) in a study for Europe. This fact is problematic because it causes the omitted variable problem which may bias the regressions estimates. Some panel data approaches partially deal with this problem by including fixed effects which might control for this kind of omitted variable (Islam, 1995). Despite such omission, the neoclassical growth framework (see Solow, 1956) provides a simple rationale for the convergence hypothesis. The convergence property comes from the law of decreasing returns to capital accumulation [i.e., capital tends to accumulate slower (faster) in regions where it is relatively abundant (scarce)]. We introduce the initial level of income to control for decreasing returns to capital accumulation (Ottaviano and Pinelli, 2006). It is important to explain why the magnitudes of the convergence coefficient are expected to be different depending on the spatial scale. In the regressions estimates, the sizes of the initial income per capita coefficients are expected to be larger at finer spatial scales because, for instance, municipalities resemble the notion of an open economy with perfect capital mobility. Barro et al.'s (1995) neoclassical model of open economy with perfect capital mobility predicts (the possibility) that economies will jump instantaneously to a steady state of income per capita, which can be understood as a higher rate of convergence. The assumption of a more open economy is not difficult to justify in the municipal level context in Brazil, considering that the intensity of flows of capital, trade and people across municipal borders is higher than across state borders. States can thus be viewed as more closed economies than municipalities.
- 2) Human capital. Glaeser *et al.* (2003) show that the presence of positive spillovers or strategic complementarities creates a "social multiplier" where aggregate coefficients of human capital (proxied, for instance, by years of schooling) will be greater than individual coefficients. In the context of this current study, we can think of municipalities as being the micro (individual) level of analysis. For this reason,

it is possible to argue that at the aggregate level (e.g., at micro-regional or state level), the coefficient of human capital may be inflated by externalities. Glaeser *et al.* (2003) point out that the coefficients may rise with the level of aggregation due to the existence of a social multiplier, which also supports the idea that there are human capital spillovers, as suggested by a wide body of literature (e.g. Lucas 1988 and Rauch 1993).

- 3) Population density. New economic geography models (Baldwin and Forslid, 2000) show the positive impact of agglomeration externalities on economic growth rates. Population density is expected to be the proxy for agglomeration effects within a region. The magnitude of these agglomeration effects may depend on the spatial scale of analysis, because, for instance, population density probably appears to be higher at a finer scale (e.g., municipalities) than population density at a spatial scale within larger regions (such as a state). Thus, centripetal effect of agglomerations might be operating at finer scales or, in other words, agglomeration-related centripetal forces may be much more relevant at the local than at the state level.
- 4) Transportation costs. Theoretical models (Ottaviano and Puga, 1998; Lafourcade and Thisse, 2008) have shown that with decreasing transportation costs, regional inequalities will increase and then decrease. Other models integrate an endogenous growth model with the core-periphery model showing that a decrease in transportation costs may have non-linear effects on growth (Baldwin *et al.*, 2003). For the Brazilian case, Da Mata *et al.* (2007a) have found that transportation costs are inversely related to the rate of economic growth. The impact of transportation infrastructure on economic growth may vary as the geographic scale of analysis changes. For instance, if this impact is analysed at the state level, the focus will be on the connectivity between these aggregate regions. On the other hand, at the municipal level, such an analysis might examine the impact of transportation cost reductions within the borders of states.
- 5) Population growth. Barro and Sala-i-Martin (2003) point out that population growth represents the behavior of fertility, mortality and migration. The impacts of population growth on economic growth may display different results across different geographic scales, because migration pattern which is one component of population growth may vary across different scale levels (e.g., intra- versus interregional migration). For instance, the contrast in area sizes means that daytime commuting across municipalities can be more significant if compared to states. Moreover, if we are able to analyze only the migration effects, we need to bear in mind that, unlike newly born persons, migrants come with accumulated human capital; and for this reason, the results depends on whether immigrants have more or less human capital (i.e., if they are typically skilled or unskilled) than the residents of the receiving region (Barro and Sala-i-Martin, 2003).

Evaluating Multiple Spatial Dimensions of Economic Growth in Brazil using Spatial Panel Data Models (1970-2000)

Discussion Paper

> 6) Spatial externalities. The spatial growth model specifications discussed in the next section seek to capture the effects of spatial externalities. It is important to note that the extent and strength of these externalities may depend on the level of aggregation of the spatial units. For instance, spatial autocorrelation might be higher at the municipal level than at the state level, because, for instance, states are more self-contained than municipalities or, in other words, states are much more closed economic entities than municipalities. As noted by Oates (1999) it is possible to increase the size of the jurisdiction to deal with such spillovers, thereby internalizing the benefits and costs. Corrado and Fingleton (2012) note that hierarchical models (also known as multilevel models) can be used in regional science and spatial economics to study a hierarchy of effects from cities, regions containing cities, and countries containing regions; thus, not recognizing these effects emanating from different hierarchical levels can lead to incorrect inference. Herein, the adopted approach is to systematically replicate the regression specification chosen to examine the extent of spatial externalities at a single scale across multiple spatial scales. Lall and Shalizi (2003) enumerate some theoretical reasons why location effects and spatial externalities matter in examining determinants of growth that include: *i*) agglomeration economies;<sup>5</sup> *ii*) Marshallian externalities of knowledge diffusion and labor market pooling;6 *iii*) common informal norms and institutions;<sup>7</sup> iv) policy adoption.<sup>8</sup> Although much of these theoretical arguments on spatial externalities are about their positive effects, it is possible to point out some reasons why negative spatial externalities effects can be observed. For instance, with regard to the policy adoption argument discussed above, there could also "be negative policy imitation where governments may not necessarily maximize

growth but maximize rent-seeking and this behavior may be imitated by governments

<sup>5. &</sup>quot;Drawing on the central argument of the 'new economic geography' literature, growth in any region is influenced by its ability to access large markets (Krugman, 1995; Venables, 1998). These economies are not a function of the size of a specific industry but of the overall size of the agglomeration. Thus, competitive enterprises accessing larger markets can enhance productivity. In addition to market size, agglomeration benefits potentially include access to specialized services (banking and finance), interindustry linkages, physical and economic infrastructure, and a larger medium for information exchange. Limiting the scope of the analysis to administrative units without considering the economic agglomeration (to which the region may belong) and the effects of market access are likely to limit the scope of the analysis" (Lall and Shalizi, 2003: 664).
6. "For technological externalities, innovations in one region are adopted in neighboring regions through diffusion, thereby creating convergence in production processes and linkages in development outcomes. In Marshall's (1920) terminology, diffusion is spatially localized and does not extend to all locations. The second source of Marshallian agglomeration is labor market pooling, where production units in one region can gain access to a shared pool of labor in the larger regional economy" (Lall and Shalizi, 2003: 664).

<sup>7. &</sup>quot;Neighboring regions are quite likely to share common informal norms and institutional structures making them react similarly to exogenous shocks (North, 1990)" (Lall and Shalizi, 2003: 665).

<sup>8. &</sup>quot;Growth rates could be correlated across space due to 'copycat policy adoption' (Easterly and Levine, 1998). They suggest that policies leading to high growth may provide a model of the efficacy of public intervention to governments in neighboring regions" (Lall and Shalizi, 2003: 665).

*in neighboring regions*" (Lall and Shalizi, 2003: 665). Moreover, Lall and Shalizi (2003: 679) suggest that improvements in the structural variables (e.g., economic structure, workforce quality, and infrastructure quality) are likely to increase growth performance in the region; however, "*if growth in a particular region is higher than that of its neighbors, the region is likely to attract mobile capital and skilled labor from neighboring regions, thereby having a detrimental effect on growth performance in neighboring regions*". The spatial models discussed next are a way to model these spatial externalities via spatial lags of the dependent and/or explanatory variables and/or spatial disturbances specifications.

## **3 SPATIAL PANEL DATA MODELS**

To study the impact of the explanatory variables upon economic growth at a variety of geographic scales, we employed spatial panel data models. These models try to account for spatial correlations, allowing at the same time for the existence of idiosyncratic effects (fixed or random) for the regional observational units. Indeed, the presence of spatially autocorrelated residuals in the non-spatial growth regressions motivates the estimation of the spatially augmented Solow models – as presented in Rey and Montouri (1999), López-Bazo *et al.* (2004) and Ertur and Koch (2007), for instance – to deal with such spatial autocorrelation.

However, it is worth noting that there are alternative explanations for the existence of spatial autocorrelation in the residuals of the growth equations. Nuisance spatial dependence (spatial error) is one potential justification. As explained by Magrini (2004: 2763), it "may result from measurement problems such as a mismatch between the spatial pattern of the process under study and the boundaries of the observational units." It is also likely that regions that are geographically close together may experience random shocks that affect both simultaneously. Another explanation is related to unobserved determinants that are correlated across regions (Finglenton and López-Bazo, 2006). Possible unobserved determinants of economic growth not considered in these models include cultural, institutional and technological factors, which might be correlated across spatial units.

Moreover, Fingleton and López-Bazo (2006) note that substantive dependence (spatial lag and/or spatial cross-regressive) assumes that across-region externalities are due to knowledge diffusion and pecuniary externalities. López-Bazo *et al.* (2004) discuss in some detail the substantive arguments for spatial dependence across regions. These authors built a spatially augmented growth model based on Mankiw *et al.* (1992) demonstrating that economic growth and initial productivity in the other regions boost growth in a given

region, which is explained by regional spillovers of the diffusion of technology from other regions, caused by investments in physical and human capital. However, López-Bazo *et al.* (2004) recognize that it is also plausible that these externalities across economies might be caused by pecuniary externalities other than knowledge spillovers – such as those created by a specialized market for labor or output, or forward and backward linkages drawn from trade in intermediate goods – which are related with increasing returns at the firm level, as noted by contributors from the so-called new economic geography (Fujita *et al.*, 1999).

Accounting for these spatial autocorrelations in growth regressions is essential to have reliable inferences. Of note, Baltagi and Pirotte (2010) examine the standard panel data estimators under spatial dependence using Monte Carlo experiments and show that when the spatial coefficients are large, hypothesis test(s) based on standard panel data estimators that ignore spatial dependence can lead to misleading inference. Moreover, Arbia and Petrarca (2011) present a general framework to investigate the effects of MAUP on spatial econometric models showing how the presence of spatial effects affects the classical results. Arbia and Petrarca (2011) concentrate on the loss in efficiency of the parameters' estimators due to aggregation.

Recently, new developments on spatial panel data models have emerged in the spatial econometrics liter ature, proposing alternative spatio-temporal models to investigate convergence and growth of regions (Elhorst *et al.*, 2010), regional markets (Keller and Shiue, 2007), and labor economics (Foote, 2007), among other fields. Anselin *et al.* (2008) provide a list of alternative spatial panel data models. In the same way, Elhorst (2012: 5) examines a collection of spatial dynamic panel data (SDPD) models that include one or more of the following variables and/or error terms: "*a dependent variable lagged in time, a dependent variable lagged in space, a dependent variable lagged in space, serial error autocorrelation, spatial error autocorrelation, spatial-specific and time-period-specific effects*".

Lee and Yu (2010) examine some recent developments in spatial panel data models for both static and dynamic cases which consider the fixed effects, spatial lags and spatial disturbances specifications [for other surveys, see also Elhorst (2010a, 2012)]. Specifically, these spatial dynamic panel data models can be applied to investigate economic growth and convergence processes of regions which employ income per capita growth rates versus lagged levels of the explanatory variables. To study the robustness of the results, we considered several model specifications for the spatial dependence structure. Both fixed and random effect models were estimated, considering spatial lags for the dependent and explanatory variables, and considering different error spatial dependences. The general model specification is given by

$$y = \lambda (I_T \otimes W) y + X\beta + u, \tag{1}$$

where *y* is the vector with the dependent variable, *X* is the matrix with explanatory variables (which includes the log of the initial period level of the dependent variable),  $\beta$  is the vector of unknown coefficients, and *W* is the weight matrix built according to the neighborhood relations among observational units. Coefficient  $\lambda$  corresponds to the autoregressive parameter, and it gives a measure of the spatial dependence between the response variable in different geographic units. We assume that  $\lambda$  is smaller than one in absolute value.

As normally used in spatial econometric models, the *W* matrix used in this paper is standardized, such that all rows sum to one (see Anselin, 1988). The term  $l_t$  corresponds to the identity matrix, with dimension *T* (the number of time periods), and symbol  $\otimes$  corresponds to the Kronecker product. The matrix representation above assumes the data are sorted such that the first block of observations corresponds to the first time period, the second block corresponds to the second time period, and so on.

The error term u can have several representations, varying according to the presence of idiosyncratic variability among regional units, and varying according to the spatial dependence structure. The discussion here considers only models with fixed or random effects, for models with none of these effects can be easily derived from the ones presented below. Besides, the several specification tests performed on our empirical data rejected the hypothesis of no idiosyncratic terms.

Assuming that there are idiosyncratic effects, represented by the terms in the N dimensional vector  $\mu$  (where N is the number of observational units), the first specification considered for the term u is given by:

$$\mathbf{u} = (\mathbf{i}_T \otimes \mathbf{I}_N) \boldsymbol{\mu} + \boldsymbol{\epsilon}, \tag{2}$$

where  $i_t$  is a  $T \times 1$  vector of ones,  $I_N$  is an identity matrix with dimension N, and  $\epsilon$  is a vector with dimension  $TN \times 1$ , with the error terms. Spatial dependence can also be assumed for a  $\epsilon$  ccording to

$$\epsilon = \rho(I_T \otimes W) \epsilon + \nu,$$
 (3)

where  $\rho$  is an autoregressive coefficient for the error terms (we assume that  $\rho$  is smaller than one in absolute value) and  $\nu$  is a random vector, with all terms independent and normally distributed with zero mean and variance  $\sigma_{\nu}^{2}$ . We will refer to the representation in (2) as Baltagi's representation (see Baltagi, 2008, and Baltagi *et al.*, 2003).

Another way to include spatial dependence along with idiosyncratic terms in u the term in equation (1) is by specifying the spatial lags directly for the term u, according to

$$u = \rho(I_T \otimes W)u + \epsilon, \tag{4}$$

with  $\epsilon = (i_T \otimes I_N) \mu + \nu$ . This representation was considered in Kapoor *et al.* (2007). Because of the structure in (4), spatial correlation applies to both the individual effects in and the remainder error terms in  $\epsilon$ . We will refer to this second model as Kapoor's representation. Estimation of models including both the spatial lag term  $(I_T \otimes W)y$  and spatial dependence for the error components with Kapoor's representation is possible only when using random effects estimation. For fixed effects, one can only use Baltagi's representation.

Finally, it is important to note that Gibbons and Overman (2012) criticize these spatial models by arguing that distinguishing which of these spatial models generates the data that the researcher has at hand is very difficult in applied research. For instance, it is hard to discriminate the model implied in Equation (1) from the model represented in Equation (5):

$$y = X\beta + \delta_x (I_T \otimes W) X + u \,. \tag{5}$$

Equation (5) describes the spatial model known as spatially lagged X regression model (SLX, or spatial cross-regressive model) that assumes interactions between exogenous characteristics of nearby observations (WX) directly affect Y. Gibbons and Overman (2012) point out that researchers interested in spatial spillovers should incorporate a reduced form specification such as in Equation (5), which may better identify causality in most cases (see Partridge *et al.*, 2012, for a comprehensive discussion).<sup>9</sup> Equation (5) may incorporate a spatial lag of the dependent variable

<sup>9.</sup> Partridge *et al.* (2012: 170) further explain that "Gibbons and Overman argue that their preferred starting point is natural experiments that use geographical, institutional, or historic factors to identify causality. While natural experimental approaches raise their own problems of only assessing "cute" experiments or searching for valid instruments, Gibbons and Overman argue that identification is more transparent and less prone to the errors we described above".

(known as spatial Durbin model/SDM) or spatial dependence for the error components (known as spatial Durbin error model/SDEM). The empirical exercise conducted herein compares alternative spatial panel data growth models across a variety of geographic scale. The purpose of this approach is to better understand the determinants of Brazilian economic growth and their respective spatial spillovers (because policy conclusions greatly differ across alternative models) as well as to show that the geographic scale of analysis is an unavoidable feature (because growth determinants would differ across different spatial scales). All estimations presented in this paper were performed using the splm package in R (see Millo and Piras, 2012).

### 4 DATA

FIGURE 1

In order to evaluate the results of regional economic growth estimates at a variety of geographic scales, this paper applied alternative spatio-temporal models to the dataset used in Resende (2013). Figure 1 presents the four Brazilian geographic stratifications in the dataset – 27 states, 134 meso-regions, 522 micro-regions and 3,657<sup>10</sup> minimum comparable areas (MCAs) – and shows some statistics concerning their sizes (in square kilometers). The data are drawn from the MCA level, which is the most disaggregated spatial units in this study, and then grouped to form the other spatial scales.



Note: Own elaboration from data of IBGE. \* Minimum Comparable Areas (MCAs).

<sup>10.</sup> The total number of MCAs is 3,659, but this paper uses 3,657. Fernando de Noronha (in the state of Pernambuco) and Ilhabela (in the state of São Paulo) were excluded because they are islands and do not adjust to the spatial weight matrices used in the analyses. These exclusions do not alter the results of the paper.

MCAs were defined by Reis et al. (2005) as sets of municipalities whose borders were constant from 1970 to 2000, to address the comparability problem generated by the increase in the number of municipalities from 3,951 in 1970 to 5,507 in 2000. Brazil is divided into 27 states<sup>11</sup> that are the main political-administrative units in the country. Municipalities (MCAs in the case of this paper) represent the smallest administrative level, dealing with local policy implementation and management. Micro- and meso-regions are homogeneous regions defined by IBGE (Brazilian Institute of Geography and Statistics - Instituto Brasileiro de Geografia e Estatística) as a group of contiguous municipalities within the same state. Micro-regions were grouped according to natural and production characteristics. Meso-regions are larger areas than micro-regions and were put together according to the following dimensions: social aspects, natural setting, and communication network as an element of space articulation. According to IBGE (2011) the division of Brazil into micro- and meso-regions is "relevant to formulate public policies; to subsidize the system of decisions relative to the localization of economic, social and tributary activities; to subsidize the planning, surveys and identification of space structures of metropolitan areas and other forms of urban and rural agglomerations."

Data were collected from IPEADATA (Institute for Applied Economic Research – Instituto de Pesquisa Econômica Aplicada/IPEA), which has organized the population census information (from IBGE) of 1970, 1980, 1991 and 2000. Based on these four data points, the dependent variable was calculated as the average annual income per capita growth rate<sup>12</sup> for each time span: 1970-1980, 1980-1991 and 1991-2000. Per capita income information is deflated to Real (R\$, the Brazilian currency) in 2000.

Table 1 presents summary statistics for the dependent variable for each of the three time periods at the different spatial scales. For each decade, the simple mean of the averaged annual income per capita growth rates increases in absolute value with the level of disaggregation. In the first period (1970-1980), while the average of income per capita growth rates was 8.81% at the state level, it was 9.36% at the MCA level. Similarly, in the period between 1980 and 1991, the fall in income per capita was more intense at the MCA level (-1.71%) than at the state level (-0.72%). The same pattern

<sup>11.</sup> More precisely, there are 26 states and one federal district.

<sup>12.</sup> The income per capita growth rates are averaged over ten years because MCA data are only available from the Brazilian population censuses conducted every ten years. Furthermore, given the presence of business cycle effects, the choice of ten-year growth averages seems to be a reasonable approach to avoid those influences (Caselli *et al.*, 1996). For instance, the 1973 and 1979 "oil price shocks" affected the Brazilian economy. In 1994, Brazil launched the "Plano Real" (Real Plan), the stabilization program that ended a long period of high inflation rates that had started in the 1970s.

occurs between 1991 and 2000, when the average growth of income per capita reached 6.83% at the MCA level against 6.10% at the municipal level. Data dispersion is also higher in the more disaggregated geographic scales.

Summary stati	istics of income p	er capita growth rat	les variable	
Spatial Scales	States $(n = 27)$	Meso-regions $(n = 134)$	Micro-regions (n = 522)	MCAs* (n = 2,657)
	Avera	ged annual per capita growt	th rates	
		1970-1980		
Mean	0.0881	0.0918	0.0915	0.0936
Minimum	0.0474	0.0474	0.0140	-0.0938
Maximum	0.1088	0.1329	0.1437	0.3709
Standard deviation	0.0152	0.0157	0.0191	0.0296
		1980-1991		
Mean	-0.0072	-0.0134	-0.0154	-0.0171
Minimum	-0.0266	-0.0756	-0.0786	-0.1297
Maximum	0.0143	0.0163	0.0329	0.2190
Standard deviation	0.0100	0.0125	0.0139	0.0232
		1991-2000		
Mean	0.0610	0.0641	0.0658	0.0683
Minimum	0.0118	0.0118	-0.0007	-0.1934
Maximum	0.0766	0.0994	0.1558	0.1991
Standard deviation	0.0148	0.0135	0.0160	0.0252
	Average numbe	r of regions using the queen	contiguity matrix	
Queen (W matrix)	3.8	5.1	5.6	5.9

TABLE 1 Summary statistics of income per capita growth rates variable

Note: Own elaboration from data of IBGE. \* Minimum Comparable Areas (MCAs).

Figure A.1 (in Appendix A) maps the income per capita in 1970, 1980, 1991 and 2000 across minimum comparable areas (MCAs), micro-regions, meso-regions and states. These maps across time and geographic scales are an interesting way to visualize the dynamics of the income per capita presented in Table 1. Beyond the clear pattern of spatial concentration of income per capita across space in Brazil, it is possible to observe the drop of income per capita between 1980 and 1991. This period is known as the Brazilian "lost decade", which was an epoch of debt crises, hyperinflation and high rates of unemployment (Baer, 2003).

Explanatory variables are given in terms of initial values, that is, values in 1970, 1980 and 1991. The socioeconomic data are logged per capita income, logged average years of schooling, logged population density and population growth.<sup>13</sup> Logged

<sup>13.</sup> Population growth is adjusted for depreciation (d) and technological growth (g), under the usual assumption that d+g equals 0.05 (e.g., Mankiw *et al.*, 1992). The natural log of this variable is not taken due to the presence of negative values.

transportation costs between MCAs and São Paulo city are also from IPEADATA. The cost of transportation to São Paulo is calculated through a linear program procedure as the minimum cost (given road and vehicle conditions) of travelling between a MCA's major headquarters and São Paulo. These transportation cost data are available for the years 1968, 1980 and 1995. Values for the years 1970 and 1991 were estimated via interpolation. Summary statistics of these variables are presented in the appendix A (table A.1). Finally, the econometric specifications include time dummies for the decades of 1980 and 1990 (the time dummy for the 1970 decade was excluded from the regressions to avoid perfect multicollinearity).

The spatial weight (W) matrix used herein is the standardized first-order contiguity matrix (also called the queen contiguity matrix), in which the element  $w_{ij}$  in the matrix is 1 if areas *i* and *j* share borders or vertices, and 0 otherwise. The average number of regions using the queen contiguity matrix is presented in the bottom part of table 1. Moreover, *k*-nearest neighbors weight matrices (in which each region has the same number of neighbors) were used (k=5 and k=10) for robustness checks and the main qualitative results remain the same.

# **5 EMPIRICAL RESULTS**

In this section, we present the estimation results for models estimated at different Brazilian geographic scales. Initially, we present the results for the model estimated at the less aggregated level, which corresponds to the Minimum Comparable Areas (MCAs). The following subsections present the results for estimation at increasingly aggregated levels: microregions, mesoregions and states. Discussions and comparisons of the results across the geographic scales are conducted in Section 6.

5.1 Results at MCA Level

Initially, we employed several Lagrange multiplier tests for panel data models, to test for the presence of spatial correlation in the observations and to test for the presence of idiosyncratic effects (individual regional effects). These tests are discussed in Baltagi *et al.* (2003) and Millo and Piras (2012). A summary of the tests results are presented in table 2 below. Tests for the presence of individual regional effects are based on random effects models, such that the null hypothesis contains an assumption that  $\sigma_u^2 = 0$ .

If the null hypothesis is rejected, there is evidence against using a pooled regression estimator or some variation of it that accounts for spatial dependence.

TABLE 2 Lagrange multiplier tests for the presence of regional individual effects and the presence of spatial correlation at the MCA level

Test #	Hypothesis tested	P-value
1	Null hypothesis of $\lambda = 0$ and $\sigma_{\mu}^2 = 0$ , under the alternative that at least one of these two components is not zero	<0.0000
2	Null hypothesis of $\sigma_{\mu}^2 = 0$ , assuming $\lambda = 0$ , under the one-sided alternative that $\sigma_{\mu}^2 > 0$	
3	Standardized version of test (2) above	
4	Null hypothesis of $\lambda = 0$ , assuming that $\sigma_{\mu}^2 = 0$ , under the two-sided alternative that $\lambda \neq 0$	< 0.0000
5	Standardized version of test (4) above	0.9992
6	Null hypothesis of $\sigma_{\mu}^2$ = 0, assuming possible existence of spatial dependence ( $\lambda$ may be different than 0), under the one-sided alternative that $\sigma_{\mu}^2 > 0$	<0.0000
7	Null hypothesis of $\lambda = 0$ , assuming possible existence of random effects ( $\sigma_{\mu}^2$ may be different that 0), under the two-sided alternative that $\lambda \neq 0$	<0.0000

Note: Own elaboration.

The results in table 2 above indicate the presence of both regional individual effects and spatial correlation in the panel regression model. For tests (2) and (3), the resulting tests statistics had negative values, which may be due to the fact that these tests assume the absence of spatial correlation. In the same way, test (5) had a resulting *p*-value very close to one, although the unstandardized version resulted in a *p*-value close to zero. In any case, tests (2), (3), (4) and (5) suffer of lack of robustness, because they assume either absence of spatial correlation or absence of regional individual effects. Tests (6) and (7) are more robust, and they both indicated the presence of both effects (spatial correlation and idiosyncratic effects).

Based on the test results, we can proceed to parameter estimation. We estimated both fixed effects and random effects models, so as to have an idea of the robustness of our conclusions. Tables 3 and 4 below show the estimation results for models estimated at the MCAs level. Table 3 has results for fixed effects models, whereas table 4 presents the results for random effect models. For random effects, we also estimated models with Kapoor's representation (not shown), but the results were similar to the results from Baltagi's representation.

_	Spatial dependence specifications				
Variable	Non-spatial dependence (1)	Spatial lag (2)	Baltagi's representation (3)	Spatial lag and spatial error dependence (4)	
Log per capita income	-0.105100***	-0.104850***	-0.106238***	-0.106348***	
Log population growth $(n+g+d)$	0.024271	0.027144	0.030318*	0.029922	
Log education	0.000949*	0.000908*	0.001230***	0.001387**	
Log transportation cost to São Paulo	0.005549	0.004522	0.006898**	0.008469**	
Log population density	-0.008891***	-0.008674***	-0.007647***	-0.007279***	
Dummy variable for decade 1980	-0.009935***	-0.005518***	-0.008591***	-0.014813***	
Dummy variable for decade 1991	0.056904***	0.057344***	0.057767***	0.056787***	
Spatial lag for the dep. variable ( $\lambda$ )		0.045498***		-0.061589**	
Spatial lag for error components ( $ ho$ )			0.183343***	0.270577++	
<i>R</i> -squared	0.926923	0.927062	0.926889	+++	

### TABLE 3 Estimation results for spatial panel data fixed effects models at the MCA level

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%. (\*) means that there was no standard error or *p*-value reported, and (\*) means that the residuals do not sum to zero and the *R*-squared was not calculated. Model (1), with no spatial dependence, was estimated using the within estimator. Models for spatial lag (2) and for Baltagi's representation (3) were estimated by maximum likelihood. Model (4), including both spatial dand error spatial dependence, was estimated using GMM.

According to the results in table 3, we note that the coefficients for spatial dependence ( $\lambda$  and  $\rho$ ) are statistically significant for all fixed effects specifications. The estimates for  $\rho$  seem to be higher than the estimates for  $\lambda$ . When both spatial parameters are included (model 4), the autocorrelation parameter  $\lambda$  has a negative sign. When spatial dependence for the error terms (models 3 and 4) is included, transportation cost to São Paulo becomes statistically significant. Moreover, from the analysis of models 2, 3 and 4, higher economic growth rates at the MCA level are positively related to education and negatively associated with income per capita (conditional convergence) and population density. It is important to note the statistical significance of the time dummies for the 80s and 90s. Of note, high *R*-squared values can be observed in all estimations. For instance, the R-squared in column 1 (non-spatial model at the AMC level) is 0.9269. However, if the time dummies are dropped from the regression the R-squared goes to 0.5720 (not shown in table 3). This means that time dynamics have a relevant explanatory power in the Brazilian case. This fact is observed for all estimation techniques and geographic scales.

		Spatial depender	nce specifications	
Variable	Non-spatial dependence (5)	Spatial lag (6)	Baltagi's representation (7)	Spatial lag and Baltagi's representation (8)
Intercept	0.722365***	0.281552***	0.296934***	0.327859***
Log per capita income	-0.095229***	-0.028705***	-0.029159***	-0.027410***
Log population growth $(n+g+d)$	0.067672***	-0.010857	-0.008534	0.001988
Log education	0.003236***	0.008781***	0.008672***	0.007427***
Log transportation cost to São Paulo	-0.037963***	-0.012662***	-0.012807***	-0.011622***
Log population density	-0.000784	-0.000583**	-0.000562**	-0.000441*
Dummy variable for year 1980	-0.037403***	-0.077489***	-0.092234***	-0.148397***
Dummy variable for year 1991	0.024934***	-0.014743***	-0.017862***	-0.030344***
Spatial lag for the dep. variable ( $\lambda$ )		0.137026***		-0.502051***
Spatial lag for error components ( $ ho$ )			0.153120***	0.555498***
<i>R</i> -squared	0.912423	0.774965	0.804480	0.411778

TABLE 4
Estimation results for spatial panel data random effects models at the MCA leve

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%. Models (5), (6), (7) and (8) were estimated by maximum likelihood. For models (7) and (8), using a Baltagi's representation for error components spatial dependence, we estimated equivalent models, with Kapoor's representation, and the results were very similar.

From table 4, we notice that the results for the fixed effects and for the random effects models are quite similar. The random effects models assume that the regional individual effects are uncorrelated with the error terms, whereas for the fixed effect models, this correlation needs not be null. To test whether the zero correlation assumption is valid, so that we can rely on the random effects estimation results, we can resort to the spatial Hausman test, discussed in Mutl and Pfaffermayr (2011). The results for the spatial Hausman test, considering an underlying model with both spatial lag ( $\lambda \neq 0$ ) and error spatial dependence ( $\rho \neq 0$ ), rejected the null hypothesis that both fixed effects and random effects models are equivalent, with a *p*-value smaller than 1.0 and -10. Therefore, we have evidence in favor of considering the results from the fixed effects model. Indeed, Durlauf *et al.* (2005) note that most of panel data growth studies employ a fixed effects (within-group) estimator rather than a random effects terms across the four geographic scales under investigation.

<sup>14.</sup> Durlauf *et al.* (2005) explain that standard random effects estimators require that the individual effects are distributed independently of the explanatory variables, and this requirement is clearly violated for a dynamic panel by construction, given the dependence of log initial period level of the dependent variable on individual effects.

To better understand the variables affecting municipality growth, we present in table 5 the estimation results for a fixed effects model, where we add spatial lags for the explanatory variables (*WX*) to the list of right-hand-side variables. We estimate the model with and without a spatial lag. Note that the model with a spatial lag (10) is known as the Spatial Durbin Model (SDM). LeSage and Fischer (2008) prefer the SDM specification arguing that the conjunction of plausible circumstances likely to arise in applied spatial growth regression modeling makes this model specification a natural choice over competing alternatives. However, Gibbons and Overman (2012) favor model (9). For identification reasons, the model with both spatial lag and spatial error dependence could not be estimated by maximum likelihood.<sup>15</sup> Numerical instability also occurred when using generalized method of moments for the model with both spatial components, because the instruments contain spatial lags of independent variables, and these lags are already included into the right-hand-side variables. Therefore, the specifications in table 5 either contain a spatial component in the error terms or include a spatial lag for the dependent variable.

Note that the three models estimated in table 5 present similar results. Higher economic growth within one MCA is negatively related to income per capita (conditional convergence) and positively associated with population growth and education. Furthermore, as argued by Sardadvar (2012) the results show how education levels are beneficial to economic growth if found within one MCA, but disadvantageous if found in neighboring regions ( $W^*$ log education). Moreover, the results show that economic growth within one MCA is positively influenced by its neighbors' income per capita levels ( $W^*$ log per capita income).<sup>16</sup> Moreover, other coefficients for spatial dependence ( $\lambda$  and  $\rho$ ) present positive signs and are statistically significant.

<sup>15.</sup> The over parameterization, in terms of spatial components, caused numerical instability in the optimization process. 16. Sardadvar (2012) develops a spatial neoclassical growth model explaining these results.

	Spatial dependence specifications				
Variable —	Spatial cross-regressive model (SLX) (9)	SDM (10)	Baltagi's representation (SDEM) (11)		
Log per capita income	-0.11501***	-0.11496***	-0.11510***		
Log population growth $(n+g+d)$	0.04718*	0.04873**	0.04855**		
Log education	0.00133**	0.00130***	0.00147***		
Log transportation cost to São Paulo	0.00489	0.00451	0.00427		
Log population density	-0.00181	-0.00166	-0.00161		
W*log per capita income	0.03485***	0.03608***	0.03392***		
$W^*$ population growth ( $n+g+d$ )	-0.06679	-0.05918*	-0.05469		
W*log education	-0.00205*	-0.00231**	-0.00186*		
W*log transportation cost to São Paulo	0.01466*	0.01354**	0.01563**		
W*log population density	-0.01222***	-0.01167***	-0.01072***		
Dummy variable for year 1980	-0.02666***	-0.02011***	-0.02587***		
Dummy variable for year 1991	0.04759***	0.04789***	0.04779***		
Spatial lag for the dep. variable ( $\lambda$ )		0.07444***			
Spatial lag for error components ( $ ho$ )			0.15792***		

### TABLE 5 Estimation results for spatial panel data fixed effects models with spatial lags of the explanatory variables at the MCA level

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%.

### 5.2 Results at Micro-Regional Level

In total, the country is divided into 522 microregions. The results for fixed effects estimation are shown, without and with spatial lags for explanatory variables in tables 6 and 7, respectively. In table 6, only the per capita income coefficient is statistically significant for all specifications, indicating the process of conditional convergence at the micro-regional level. Moreover, in all spatial dependence specifications the coefficients for spatial dependence ( $\lambda$  and  $\rho$ ) are positive and statistically significant.

In table 7, we find that per capita income is negatively related to growth, while its spatial lag is positively associated with growth as suggested by the theoretical spatial growth models (e.g., López-Bazo *et al.*, 2004; Ertur and Koch, 2007; Sardadvar, 2012). The positive spatial lag of income per capita means that one microregion located in a relatively rich neighbourhood will tend to have a higher per capita income growth (with other things being equal). In the spatial lag model (10) population growth fosters economic growth within one microregion; but economic growth within one microregion is negatively influenced by its neighbors' population growth (*W*\*population growth). Similarly to MCAs' results, spatial dependence specifications present positive and statistically significant coefficients for spatial dependence ( $\lambda$  and  $\rho$ ); although higher values can be observed for the micro-regional level.

Estimation results for spatial panel data fixed effect models at the microregion level					
	Spatial dependence specifications				
Variable —	Non-spatial dependence (1)	Spatial lag (2)	Baltagi's representation (3)	Spatial lag and spatial error dependence (4)	
Log per capita income	-0.08506***	-0.075369***	-0.096067***	-0.095525***	
Log population growth (n+g+d)	-0.05162	-0.027045	0.061710*	0.060210	
Log education	0.00389	0.002299	0.002028	0.002161	
Log transportation cost to São Paulo	0.01059*	0.000243	0.000279	0.001400	

-0.01324\*\*\*

-0.02467\*\*\*

0.04379\*\*\*

----

0.96162

# TABLE 6Estimation results for spatial panel data fixed effect models at the microregion level

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 5%. (++) means that there was no standard error or *p*-value reported, and (+++) means that the residuals do not sum to zero and the *R*-squared was not calculated. Model (1), with no spatial dependence, was estimated using the within estimator. Models for spatial lag (2) and for Baltagi's representation (3) were estimated by maximum likelihood. Model (4), including both spatial lag and error spatial dependence, was estimated using GMM.

-0.010268\*\*\*

0.042499\*\*\*

0.395592\*\*\*

----

0.96773

0.005377

-0.000796

-0.019650\*\*\*

0.045226\*\*\*

0.664554\*\*\*

0.95976

-0.000936

-0.025571\*\*

0.043831\*\*\*

0.657960++

+++

-0.054056

### TABLE 7

Log population density

R-squared

Dummy variable for year 1980

Dummy variable for year 1991

Spatial lag for the dep. variable ( $\lambda$ )

Spatial lag for error components ( $\rho$ )

# Estimation results for spatial panel data fixed effect models with spatial lags of the explanatory variables at the microregion level

	Spatial dependence specifications				
Variable	Spatial cross-regressive model (SLX) (9)	SDM (10)	Baltagi's representation (SDEM) (11)		
Log per capita income	-0.09654***	-0.09822***	-0.09565***		
Log population growth $(n+g+d)$	0.06554	0.08678**	0.06248*		
Log education	0.00229	0.00141	0.00181		
Log transportation cost to São Paulo	-0.00410	-0.00563	-0.00389		
Log population density	0.00133	0.00269	-0.00024		
W*log per capita income	0.02468***	0.07053***	0.01525**		
$W^*$ population growth ( $n+g+d$ )	-0.33368***	-0.17670**	-0.14330		
W*log education	-0.00003	-0.00045	0.00495		
W*log transportation cost to São Paulo	0.02379	0.01459	0.02098*		
W*log population density	-0.03024***	-0.01412***	-0.01478**		
Dummy variable for year 1980	-0.03042***	-0.01076***	-0.02777***		
Dummy variable for year 1991	0.04410***	0.01798***	0.04157***		
Spatial lag for the dep. variable ( $\lambda$ )		0.61882***			
Spatial lag for error components ( $ ho$ )			0.62415***		

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%.

### 5.3 Results at Meso-Regional Level

We now present the results using 134 Brazilian mesoregions. Tables 8 and 9 show the results for fixed effects estimations, without and with spatial lags for explanatory variables, respectively. In table 8, the per capita income coefficient is statistically significant for all specifications and indicates conditional convergence process at meso-regional level. Moreover, higher economic growth is positively related to education level for all specification in table 8. In models (2) and (3), the coefficients for spatial dependence ( $\lambda$  and  $\rho$ ) are positive and statistically significant. However, when both spatial components are included in specification (4), the statistical significance disappears.

In table 9, the spatial models (10) and (11) show that there is conditional convergence and that higher educational level is beneficial to growth within mesoregions. The coefficients of the spatial lags of population density are statistically significant in all specification and show that economic growth within one mesoregion is negatively influenced by its neighbors' population density ( $W^*$ log population density). Moreover, spatial dependence specifications present positive and statistically significant coefficients for spatial dependence ( $\lambda$  and  $\rho$ ); although their magnitudes are lower than those estimated at the micro-regional level.

	Spatial dependence specifications				
Variable	Non-spatial dependence (1)	Spatial lag (2)	Baltagi's representation (3)	Spatial lag and spatial error dependence (4)	
Log per capita income	-0.07795***	-0.07343***	-0.08412***	-0.08137***	
Log population growth $(n+g+d)$	-0.14754	-0.13980*	-0.03165	-0.07946	
Log education	0.01237*	0.01010*	0.01561**	0.01362*	
Log transportation cost to São Paulo	0.01202	0.00554	0.00936	0.00713	
Log population density	-0.01829***	-0.01522***	-0.00599	-0.01009*	
Dummy variable for year 1980	-0.03101***	-0.00045	-0.02966***	-0.01001	
Dummy variable for year 1991	0.03469***	0.03824***	0.03072***	0.03631***	
Spatial lag for the dep. variable ( $\lambda$ )		0.34579***		0.20342	
Spatial lag for error components ( $ ho$ )			0.50436***	0.30934	
R-squared	0.97035	0.97368	0.96929	+++	

TΑ	В	L	E	8	

Estimation results for spati	al panel data fixed effe	ct models at the mesoregion level
------------------------------	--------------------------	-----------------------------------

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 5%. (++) means that there was no standard error or p-value reported, and (+++) means that the residuals do not sum to zero and the R-squared was not calculated. Model (1), with no spatial dependence, was estimated using the within estimator. Models for spatial lag (2) and for Baltagi's representation (3) were estimated by maximum likelihood. Model (4), including both spatial and error spatial dependence, was estimated using GMM.

#### TABLE 9

Estimation results for spatial panel data fixed effect models with spatial lags of the explanatory variables at the mesoregion level

	Spa	atial dependence specification	ons
Variable	Spatial cross-regressive model (SLX) (9)	SDM (10)	Baltagi's representation (SDEM) (11)
Log per capita income	-0.08803***	-0.08841***	-0.08685***
Log population growth $(n+g+d)$	-0.01846	0.01085	-0.03268
Log education	0.01214	0.01341*	0.01237*
Log transportation cost to São Paulo	-0.00010	0.00261	0.00326
Log population density	-0.00203	0.00062	-0.00364
W*log per capita income	0.01515	0.04256***	0.00507
$W^*$ population growth ( $n+g+d$ )	-0.44103*	-0.28020	-0.34559*
W*log education	-0.00477	-0.00829	-0.00187
W*log transportation cost to São Paulo	0.01893	0.00910	0.01534
W*log population density	-0.04230***	-0.02823***	-0.03636***
Dummy variable for year 1980	-0.02632**	-0.01022	-0.01992*
Dummy variable for year 1991	0.04734***	0.03099***	0.05006***
Spatial lag for the dep. variable ( $\lambda$ )		0.43106***	
Spatial lag for error components ( $ ho$ )			0.42779***

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%.

### 5.4 Results at State Level

Tables 10 and 11 bring the estimation results from models without and with spatial lags for explanatory variables. From table 10 we note that growth estimations at the state level using spatial dependence specifications show weak evidence of spatial spillovers. Indeed, Resende (2012) demonstrates that the diagnostics for spatial autocorrelation in the error terms using Moran's I statistics in non-spatial panel models (similar to that in column 1 in table 10) are not statistically significant at the state level in Brazil. For this reason the coefficient for spatial dependence ( $\lambda$ ) is not statistically significant in column (2) and (4); and the estimate for  $\rho$  seems to be statistically significant only at 5% level in column (3). This finding indicates that, at the state level, the use of spatial econometrics might not be necessary because the spatial autocorrelation does not appear in the residuals. In this sense, model (1) is the most appropriate to investigate economic growth determinants at the state level. In model (1), only income per capita and transportation cost coefficients are statistically significant. The former suggests conditional convergence at the state level and the latter indicates that reductions in transportation costs may have a negative impact on economic growth. Of note, as argued by Resende (2013) these results should be interpreted with caution because these estimates control for fixed effects when transportation costs already have a clear component that is fixed (namely, the distance between each spatial unit and the spatial unit represented by São Paulo). Therefore, the transportation cost coefficients might only be picking up the variable part of transportation costs (for instance, road conditions or quality) showing that improvements in road conditions are positively related to economic growth at state level. Table 11 also shows negligible effects of spatial dependence components ( $\lambda$  and  $\rho$ ) in columns (10) and (11). The next section presents a discussion of the results described herein comparing the results found across the four geographic scales.

Estimation results 1	for spatial panel d	ata fixed effect i	models at the sta	le level
		Spatial depende	nce specifications	
Variable —	Non-spatial dependence (1)	Spatial lag (2)	Baltagi's representation (3)	Spatial lag and spatial error dependence (4)
Log per capita income	-0.09311***	-0.09205***	-0.09505***	-0.09280***
Log population growth $(n+g+d)$	-0.19468	-0.19199	-0.11385	-0.15366
Log education	0.02419	0.023889*	0.03543**	0.02848*
Log transportation cost to São Paulo	0.04676**	0.043943***	0.04254**	0.04101*
Log population density	-0.01412	-0.01334*	-0.00530	-0.00911
Dummy variable for year 1980	-0.00511	0.00054	-0.01167	0.00140
Dummy variable for year 1991	0.05578**	0.05557***	0.04163**	0.05005**
Spatial lag for the dep. variable ( $\lambda$ )		0.07659		0.11809
Spatial lag for error components ( $ ho$ )			0.27317*	0.12471
<i>R</i> -squared	0.97806	0.97818	0.97723	+++

TABLE 10 Estimation results for spatial panel data fixed effect models at the state level

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%. (++) means that there was no standard error or *p*-value reported, and (+++) means that the residuals do not sum to zero and the R-squared was not calculated. Model (1), with no spatial dependence, was estimated using the within estimator. Models for spatial lag (2) and for Baltagi's representation (3) were estimated by maximum likelihood. Model (4), including both spatial lag and error spatial dependence, was estimated using GMM.

#### TABLE 11

# Estimation results for spatial panel data fixed effect models with spatial lags of the explanatory variables at the state level

	Spa	atial dependence specificatio	ns
Variable	Spatial cross-regressive model (SLX) (9)	SDM (10)	Baltagi's representation (SDEM) (11)
Log per capita income	-0.10120***	-0.10147***	-0.10008***
Log population growth $(n+g+d)$	-0.22026	-0.17916	-0.21387
Log education	0.05089**	0.05269***	0.04753***
Log transportation cost to São Paulo	0.03582	0.03592	0.03699
Log population density	-0.00655	-0.00272	-0.00553
W*log per capita income	0.03042	0.03870	0.02213
$W^*$ population growth ( $n+g+d$ )	-0.57569	-0.55927	-0.68470
W*log education	-0.07804*	-0.07703**	-0.06810**
W*log transportation cost to São Paulo	0.04382	0.03231	0.04093
W*log population density	-0.04415	-0.04387*	-0.05021**
Dummy variable for year 1980	0.01660	0.01967	0.02060
Dummy variable for year 1991	0.11129**	0.09985***	0.11220***
Spatial lag for the dep. variable ( $\lambda$ )		0.16543	
Spatial lag for error components ( $ ho$ )			0.17986

Note: For the significance level in the estimations in the table, (\*\*\*) means statistically significant at level 0.1%, (\*\*) means statistically significant at level 1% and (\*) means statistically significant at level 5%.

# 6 DISCUSSION

This section aims to discuss the results of economic growth regressions estimated at four geographic scales (MCAs, microregions, mesoregions and states) using alternative spatial panel data methods controlling for fixed effects. Non-spatial panel data models with fixed effects were also estimated for comparative purposes.

In the previous section we note that the coefficients of (log) initial income per capita are negative and statistically significant in all estimations and geographic scales. This negative correlation between the per capita income growth rate and the initial per capita income may suggest conditional  $\beta$ -convergence; but we need to take into account the interpretation of fixed effects in this result. Resende (2013) shows faster convergence rate in the fixed effects framework compared with the pooled OLS approach.

Of note, when the panel data with fixed effects is adopted in economic growth analyses, it creates a bridge between development economics and the neoclassical empirics of growth because this framework allows for unobservable differences in the production function which focuses attention on all the tangible and intangible factors (e.g., institutional characteristics) that may enter into its respective individual effect (Islam, 1995). Islam (1995) argues that persistent differences in technology level and, for instance, institutions are an important factor in understanding economic growth across regions; because when these variables are included in the regressions in the form of fixed effects, the convergence process occurs at a faster rate.<sup>17</sup> Then, improvements in these unobserved factors (e.g., technology levels and institutions) might produce positive effects on the region's long-run income level, including a higher transitional growth rate.

Herein, when the spatial distribution of these fixed effect terms is analyzed across the four geographic scales (see Figure A.2 in appendix A)<sup>18</sup> we observe a clustering of high values in the south, southeast and central-west of Brazil at the MCA spatial scale, for instance. This fact suggests that fixed effects are really capturing a higher level of, for example, technology and institutions in these regions, which are the most developed areas in Brazil, generating higher growth rates in the analyzed period. However, this spatial distribution of the fixed effects shows some variation across spatial scales. If we observe the fixed effect at the MCAs within each state, we clearly find such variability. For instance, we may conclude that the state of São Paulo presents a higher fixed effect - which may be interpreted as good institutions - that generates higher growth rates. However, we need to bear in mind that, within this state we have MCAs (municipalities) that present values for fixed effects as low as in MCAs in the North region (where we observe the clustering of low fixed effects values). In this sense, the analysis of this phenomenon at a variety of geographic scales presents us a better understanding of these fixed effects.

<sup>17.</sup> Islam (1995) explains the faster convergence rate in the FE framework compared with the pooled OLS approach arguing that in the latter approach (or in the framework of single cross-section regression) the technology variable, A(0), being unobservable or unmeasurable, is left out of the equation (or, incorporated in the error term): "[t]his actually creates an omitted variable problem. Since this omitted variable is correlated with the included explanatory variables, it causes the estimates of the coefficients of these variables to be biased. The direction of bias can be assessed from the standard formula for omitted variable bias. The partial correlation between A(0) and the initial value of y (income per capita) is likely to be positive, and the expected sign of the A(0) term in the full regression, (...), is also positive. Thus, the estimated coefficient of yi,t-1, is biased upward. (...) This explains why we get lower convergence rates from single cross-section regressions and pooled regressions that ignore correlated individual country effects" Islam (1995: 1147).

<sup>18.</sup> These fixed effects come from the spatial Durbin model (SDM). The spatial distribution of fixed effects using other spatial models is very similar; and for this reason they are not shown here.

Furthermore, it is important to note that the evidence of conditional convergence using the fixed effects approach may lead us to conclude that the club convergence hypothesis<sup>19</sup> cannot be ruled out. Indeed, there is growing evidence that the club convergence hypothesis is the correct one for the Brazilian case (Andrade et al., 2004; Laurini et al. 2005; Coelho and Figueirêdo, 2007, Resende, 2011). Of note, Islam (1995) points out that instead of using the panel data method, the other way to control for differences in technology and institutions is to classify regions (or countries) into similar clubs. The club convergence analysis allows for differences in the aggregate production function across groups of regions. Classifying regions into similar groups (or clubs) has been the approach adopted in some recent studies (Coelho and Figueirêdo, 2007; Cravo, 2010; Resende, 2011; Cravo and Resende, 2012). In this sense, the fixed effects findings described above are consistent with the club convergence hypothesis for the Brazilian case. Moreover, the variability of conditional  $\beta$ -convergence coefficients due to the geographic scale of analysis seems to be small using fixed effects models. Coefficients sizes range from -0.07343 (in table 8, column 2) at meso-regional level to -0.11510 (in table 5, column 11) at MCA level. It is worth noting that higher rates of convergence at the MCA level suggest that municipalities are more open economies (Barro et al., 1995) than more aggregated regions such as mesoregions. For this reason, MCAs present faster convergence rates to their own state-steady levels of income per capita.

Now we turn to the analysis of the variability of other estimated coefficients at different geographic scales. In most cases, the coefficients of population growth are statistically insignificant at 5% level. It is probably because of a balance between countervailing behavior effects of fertility, mortality, and migration that the population growth coefficient becomes statistically insignificant. The average-year-of-schooling coefficient inflates as more aggregate data is used. For instance, using the spatial lag model, the years-of-schooling coefficient is 0.000908 at the MCA level, 0.002299 at the micro-regional level, 0.01010 at the meso-regional level, and 0.023889 at the state level. This might suggest the strength of the spatial interactions across individuals and regions, a phenomenon coined as the "social multiplier" effect by Glaeser *et al.* (2003).

<sup>19.</sup> Ertur et al. (2006: 8) highlight that "the concept of club convergence is based on endogenous growth models that are characterized by the possibility of multiple, locally stable, steady state equilibria as in Azariadis and Drazen (1990). Which of these different equilibria an economy will be reaching depends on the range to which its initial conditions belong. In other words, economies converge to one another if their initial conditions are in the 'basin of attraction' of the same steady state equilibrium. When convergence clubs exist, one convergence equation should be estimated per club, corresponding to different regimes".

The analysis of spatial models results reveals that transportation costs effects are statistically significant to economic growth only at the state level. As explained earlier, in the fixed effects approach, the transportation cost coefficients might only be picking up the variable part of transportation costs (for instance, road conditions or quality). In this sense, we may conclude that improvements in road conditions are positively related to economic growth at the state level.

In some spatial panel data specifications, population density coefficients are negative and statistically significant at the MCA (table 3), micro-regional (table 6) and meso-regional (table 8) spatial scales. These results are contrary to the argument that agglomeration effects are beneficial to economic growth because the negative signs of the population density coefficients mean that higher populated areas are harmful to economic growth, demonstrating somehow that congestion effects might explain this negative effect for the analyzed period (1970-2000). Moreover, it is important to highlight that the magnitudes and statistical significance levels also vary across the geographic scales. For instance, at the state level, population density does not seem to be an important factor to growth.

Finally, the evidence collected from spatial spillovers coefficients shows that their magnitudes vary according to the spatial scale under analysis. In the spatial lag specification, the spatial lag for the dependent variable ( $\lambda$ ) presents positive and statistically significant coefficients at the MCA (0.045498), micro-regional (0.395592) and meso-regional (0.34579) levels. On the other hand, at state level such coefficient is no longer statistically significant suggesting that spatial spillovers are bounded in space as already found in Resende (2011). Moreover, as discussed earlier, the mechanisms that might explain the spatial interactions among regions may be related to nuisance or substantive arguments. The results suggest that both mechanisms may be the origin of spatial linkages observed in this empirical exercise. Some authors such as LeSage and Fischer (2008) prefer the spatial Durbin Model (SDM) specification over competing alternatives. This model can be supported by theoretical spatial growth model such as those developed by López-Bazo et al. (2004), Ertur and Koch (2007) and Sardadvar (2012). For instance, spatial Solow growth models demonstrate that regional spillovers of the diffusion of technology across regions are caused by the spatial dimension of investments in physical and human capital (López-Bazo et al., 2004). However, it is important to note that the correct identification of the most appropriate empirical spatial model is still a challenging issue to be addressed by the spatial econometric literature [see, for instance, Partridge et al. (2012) and Gibbons and Overman (2012)].

32

# 7 CONCLUSIONS

The goal of this paper was to evaluate the results of regional economic growth estimates at multiple spatial scales using spatial panel data models. The spatial scales examined were minimum comparable areas, micro-regions, meso-regions and states over the period between 1970 and 2000. Alternative spatial panel data models with fixed effects were systematically estimated across those spatial scales to demonstrate that the estimated coefficients change with the scale level.

The results show that the conclusions obtained from growth regressions are dependent on the choice of spatial scale. First, the fixed effects used in the spatial models allowed differences in the aggregate production function focusing attention on all the tangible and intangible fixed effects that underlie much of the discussion of development economics (Islam, 1995). We may conclude that improvements in unobserved fixed factors (e.g, technology levels and institutions) produce positive effects on growth rates at the four spatial levels. The spatial distribution of these fixed effect terms are clustered across space. One result of this finding is that club convergence hypothesis cannot be rejected suggesting there are differences in the convergence processes between the north and south in Brazil. However, spatial distribution of the fixed effects shows some variation across spatial scales. It is worth noting that higher rates of convergence at the MCA level were found, suggesting that municipalities are more open economies than more aggregated regions.

This paper also showed that determinants of economic growth estimated on different geographic scales change with the scale. In most cases, the coefficients of population growth are statistically insignificant at 5% level, except at the meso-regional level using the spatial lag specification. The positive average-years-of-schooling coefficient gets larger as more aggregate spatial scales are used. Moreover, transportation costs effect is positive and statistically significant to economic growth at the state level showing that improvements in road conditions are positively related to economic growth only at the state level. Population density coefficients show that higher populated areas are harmful to economic growth demonstrating somehow that congestion effects are operating at the MCA, micro-regional and meso-regional spatial scales, but their magnitudes vary across the geographic scales. Finally, the values of spatial spillovers coefficients also vary according to the spatial scale under analysis. In general, such coefficients are statistically significant at the MCA, micro-regional and meso-regional levels; but, at the state level those coefficients are no longer statistically significant suggesting that spatial spillovers are bounded in space.

### REFERENCES

Alesina, A., Rodrik, D. (1994) Distributive politics and economic growth. *Quarterly Journal of Economics*, 109: 465-490.

Andrade, E., Laurini, M., MADALOZZO, R., PEREIRA, P. (2004). Convergence clubs among Brazilian municipalities. *Economics Letters*, 83: 179–184.

Anselin, L. (1988). *Spatial Econometrics: Methods and Models*. Kluwer Academic Publisher, Dordrecht.

Anselin L, Le Gallo J, Jayet H (2008) Spatial panel econometrics. In: Mátyás L., Sevestre, P. (eds.) *The econometrics of panel data, fundamentals and recent developments in theory and practice*, Kluwer, Dordrecht: 627–662

Ávila, R. P., Monasterio, L. M. (2008) MAUP e a análise espacial: um estudo de caso para o Rio Grande do Sul. *Análise Econômica* (UFRGS), 26: 239-265.

Baltagi, B. H. (2008). *Econometric Analysis of Panel Data*. 4th edition. John Wiley & Sons, New York.

Baltagi, B. H., Pirotte, A. (2010) Panel data inference under spatial dependence. *Economic Modelling*, 27(6): 1368-1381.

Baltagi, B.H., Song, S.H., Koh, W. (2003). Testing Panel Data Regression Models with Spatial Error Correlation. *Journal of Econometrics*, 117, 123-150.

Barro, R., Sala-I Martin, X. (2003) Economic growth. Cambridge, Mass.: MIT, 2ed.

Behrens, K., Thisse, J. F. (2007) Regional economics: A new economic geography perspective. *Regional Science and Urban Economics*, 37: 457–465.

Bloom, D., Canning, D., Sevilla, J. (2004) The effect of health on economic growth: A production function approach. *World Development*, 32 (1): 1–13.

Briant, A., Combes, P. P., Lafourcade, M. (2010) Dots to boxes: Do the size and shape of spatial units jeopardize economic geography estimations? *Journal of Urban Economics*, 67: 287-302.

Brueckner, J. K. (1998) Testing for strategic interaction among local governments: The case of growth controls. *Journal of Urban Economics*, 44: 438–467.

Caselli, F., Esquivel, G., Lefort, F. (1996). Reopening the convergence debate: A new look at cross country growth empirics. *Journal of Economic Growth*, 1 (3): 363–389.

Coelho, R. L. P., Figueiredo, L. (2007) Uma análise da hipótese da convergência para os municípios Brasileiros. *Revista Brasileira de Economia*, 61: 331–352.

Cravo, T. A. (2010) SMEs and economic growth in the Brazilian micro-regions. *Papers in Regional Science*, 89: 711–734.

CRAVO, T., Resende, G. M. (2012) Economic growth in Brazil: A spatial filtering approach. *The Annals of Regional Science*, DOI: 10.1007/s00168-012-0504-6.

Durlauf, S. N., Johnson, P. A., Temple, J. R. W. (2005) Growth econometrics. In: Aghion, P., Durlauf, S. (Eds) *Handbook of Economic Growth*, Elservier.

Elhorst, J. P. (2010a) Spatial panel data models. In: Fischer, M., Getis, A. (eds) *Handbook of applied spatial analysis*. Springer, Berlin, Heidelberg and New York: 377–407.

Elhorst, J. P. (2010b) Dynamic panels with endogenous interaction effects when T is small. *Regional Science and Urban Economics*, 40: 272–282.

Elhorst, J. P. (2012) Dynamic spatial panels: models, methods, and inferences. *Journal of Geographical System*, 14: 5–28.

Elhorst J. P., Piras G., Arbia G. (2010) Growth and convergence in a multi-regional model with space-time dynamics. *Geographical Analysis*, 42(3): 338–355.

Ertur C., Koch, W. (2007) Growth, Technological interdependence and spatial externalities: theory and evidence. *Journal of Applied Econometrics*, 22: 1033–1062.

Ferreira, A., (2000). Convergence in Brazil: Recent trends and long-run prospects. *Applied Economics*, 32: 479-489.

Fingleton, B. (1999) Estimates of time to economic convergence: An analysis of regions of the European Union. *International Regional Science Review*, 22: 3-34.

Fingleton, B. (2011) The empirical performance of the NEG with reference to small areas. *Journal of Economic Geography*, 11: 267–279.

Fingleton, B., López-Bazo, E. (2006) Empirical growth models with spatial effects. *Papers in Regional Science*, 85: 177-198.

Fischer, M. (2011) A spatial Mankiw–Romer–Weil model: theory and evidence. *Annals of Regional Science*, 47(2): 419-436.

Gamboa, O. R. E. (2010) The (un)lucky neighbour: Differences in export performance across Mexico's states. *Papers in Regional Science*, 89 (4):777-799.

Gérard, M., Jayet, H., Paty, S. (2010) Tax interactions among Belgian municipalities: Do interregional differences matter? *Regional Science and Urban Economics*, 40: 336–342.

Gibbons, S., Overman, H. (2012) Mostly pointless spatial econometrics? *Journal of Regional Science*, 52: 172–191.

Glaeser, E. L., Sacerdote, B. I., Scheinkman, J. A. (2003) The social multiplier. *Journal of the European Economic Association*, 1 (2-3): 345-353.

IBGE (2011) *Division into regions*. Instituto Brasileiro de Geografia e Estatística. (available at: http:// www.ibge.gov.br/english/geociencias/geografia/default\_div\_int. shtm, accessed on 10/01/2011).

IPEADATA. Dados macroeconômicos e regionais (available at: http://www. ipeadata.gov.br).

Islam, N. (1995) Growth empirics: A Panel data approach. *Quarterly Journal of Economics*, 110 (4): 1127-1170.

Islam, N. (2003) What have we learnt from the convergence debate? *Journal of Economic Surveys*, 17: 309–362.

Jeanty, P., Partridge, M., Irwin, E. (2010) Estimation of a spatial simultaneous equation model of population migration and housing price dynamics. *Regional Science and Urban Economics*, 40: 343–352.

Kapoor, M., Kelejian, H.H., Prucha, I.R. (2007). Panel Data Model with Spatially Correlated Error Components. *Journal of Econometrics*, 140(1), 97-130.

Lall, S.V., Shalizi, Z. (2003) Location and growth in the Brazilian Northeast. *Journal of Regional Science*, 43: 663-681.

Laurini, M., Andrade, E., Valls, P., (2005). Income convergence clubs for Brazilian Municipalities: a non-parametric analysis. *Applied Economics*, 37 (18): 2099-2118.

Lee, L., Yu, J. (2010) Some recent developments in spatial panel data models. *Regional Science and Urban Economics*, 40: 255–271.

LeSage, J. P. (1999) *Spatial econometrics*. The Web Book of Regional Science, Regional Research Institute, West Virginia University, Morgantown, WV (available at: http://www.rri.wvu.edu/WebBook/LeSage/spatial/spatial.html)

Lesage, J. P., Fischer, M. M. (2008) Spatial growth regressions: Model specification, estimation and interpretation. *Spatial Economic Analysis*, 3(3):275.

LeSage, J. P., Pace, R. K. (2008) Spatial econometric modeling of origin-destination flows. *Journal of Regional Science*, 48: 941-967.

López-Bazo, E., Artís, M., Vayá, E. (2004) Regional externalities and growth: evidence from European regions. *Journal of Regional Science*, 44: 43–73.

Lucas, R. (1988) On the mechanics of economic development. *Journal of Monetary Economics*, 22 (1): 3-42.

Magrini, S. (2004) Regional (di)convergence. In: Henderson, J. V., Thisse, J. F. (eds) *Handbook* of *Regional and Urban Economics*, ed. 1, vol. 4, chapter 62: 2741-2796.

Mankiw, N. G., Romer, D., Weil, D. N. (1992) A Contribution to the empirics of economic growth, *Quarterly Journal of Economics*, 107: 407-37.

McDonald, S., Roberts, J. (2002) Growth and multiple forms of human capital in an augmented Solow model: A panel data investigation. *Economics Letters*, 74: 271-276.

McMillen, D.P. (2010) Issues in spatial data analysis. Journal of Regional Science, 50(1): 119-141.

Menon, C. (2012) The bright side of MAUP: Defining new measures of industrial agglomeration. *Papers in Regional Science*, 91 (1): 3-28.

Millo, G., Piras, G. (2012) splm: Spatial Panel Data Models in R. *Journal of Statistical Software*, 47(1), 1–38.

Mutl, J., Pfaermayr, M. (2011). The Hausman Test in a Cliff and Ord Panel Model. *Econometrics Journal*, 14, 48-76.

Openshaw, S., Taylor, P.J. (1981) The Modifiable Areal Unit Problem. In: Wrigley, N., Bennett, R. (ed) *Quantitative Geography, a British View*. London: Routledge and Kegan.

Partridge, M. D., Boarnet, M., Brakman, S., Ottaviano, G. (2012). Introduction: Whither spatial econometrics? *Journal of Regional Science*, 52 (2): 167-171.

Reis E. J., Tafner, P., Pimentel, M., Medina, M., Magalhaes K., Reiff, L. O., Serra R. V. (2005). *O PIB dos municípios brasileiros: metodologia e estimativas*, *1970-96*. Brasília: Ipea. Discussion Paper, n. 1064.

Resende, G. M. (2011) Multiple dimensions of regional economic growth: The Brazilian case, 1991-2000. *Papers in Regional Science*, 90 (3): 629-662.

Resende, G. M. (2013) Spatial Dimensions of Economic Growth in Brazil. *ISRN Economics*. Accepted for publication in 2012.

Rey, S. J., Montoury, B. D. (1999) U.S. regional income convergence: A spatial econometric perspective. *Regional Studies*, 33: 143-156.

Romer, P. M. (1986) Increasing returns and long-run growth. *Journal of Political Economy*, 94 (5): 1002-1037.

Sardadvar, S. (2012) Growth and disparities in Europe: Insights from a spatial growth model. *Papers in Regional Science*, 91(2): 257–274.

Solow, R. M. (1956) A contribution to the theory of economic growth. *Quarterly Journal of Economics*, 70: 65-94.

Yamamoto, D. (2008) Scales of regional income disparities in the USA, 1955–2003. *Journal of Economic Geography*, 8: 79-103.

IABLE A. I Summary statis	tics of the v	ariable	S											
				1970	-1980			1980-	1991			1991-	2000	
Variables	Spatial Scales	Obs.	Mean	Standard deviation	Minimum	Maximum	Mean	Standard deviation	Minimum	Maximum	Mean	Standard deviation	Minimum	Maximum
	MCAs*	3657	0.0936	0.0296	-0.0938	0.3709	-0.0171	0.0232	-0.1297	0.2190	0.0683	0.0252	-0.1934	0.1991
Averaged annual income per capit	a Micro-regions	522	0.0915	0.0191	0.0140	0.1437	-0.0154	0.0139	-0.0786	0.0329	0.0658	0.0160	-0.0007	0.1558
growth rate	Meso-regions	134	0.0918	0.0157	0.0474	0.1329	-0.0134	0.0125	-0.0756	0.0163	0.0641	0.0135	0.0118	0.0994
	States	27	0.0881	0.0152	0.0474	0.1088	-0.0072	0.0100	-0.0266	0.0143	0.0610	0.0148	0.0118	0.0766
	MCAs*	3657	3.6352	0.5470	0.2718	5.4339	4.5713	0.5883	2.2081	6.1779	4.3834	0.5929	2.8495	6.2442
(income new consta)	Micro-regions	522	3.7752	0.5474	2.5869	5.3925	4.6906	0.5831	3.3085	6.0514	4.5208	0.5824	3.2191	5.9361
الالالالالاله المارحة العاري	Meso-regions	134	3.9213	0.5198	2.9627	5.3059	4.8396	0.5394	3.7680	5.9564	4.6918	0.5436	3.7525	5.8522
	States	27	4.0589	0.4694	3.2472	5.0440	4.9395	0.4587	4.0932	5.8605	4.8607	0.4365	4.0990	5.8522
	MCAs*	3657	0.1037	1.0204	-46.0517	1.7228	0.5768	0.5863	-2.3026	1.9741	1.0604	0.4404	-0.9163	2.1748
In linese of exhaptions)	Micro-regions	522	0.2598	0.6186	-1.3856	1.5137	0.7112	0.5538	-0.8010	1.8083	1.1596	0.4336	-0.1731	2.0015
in (years or scriooling)	Meso-regions	134	0.4089	0.5659	-0.7391	1.4351	0.8612	0.5018	-0.2137	1.8083	1.2849	0.3910	0.4229	2.0015
	States	27	0.5577	0.4207	-0.1039	1.4351	1.0161	0.3474	0.4857	1.8083	1.4194	0.2473	1.0192	2.0015
	MCAs*	3657	0.0588	0.0270	-0.1197	0.3220	0.0605	0.0188	-0.0045	0.3084	0.0588	0.0161	-0.0164	0.2162
(n . c . d) non-lotion another **	Micro-regions	522	0.0681	0.0250	-0.0221	0.3220	0.0662	0.0193	0.0241	0.3084	0.0624	0.0122	0.0189	0.1173
עו + ט + ט, אטאמנוטוו אַוטאנוו	Meso-regions	134	0.0743	0.0231	0.0238	0.1986	0.0704	0.0168	0.0382	0.1455	0.0656	0.0112	0.0375	0.1058
	States	27	0.0851	0.0281	0.0596	0.1986	0.0775	0.0191	0.0593	0.1420	0.0707	0.0107	0.0581	0.1055
	MCAs*	3657	3.1207	1.2350	-3.9118	9.0931	3.2086	1.2619	-3.1645	9.3699	3.3244	1.2939	-2.4070	9.4510
h (nonitation donaita)	Micro-regions	522	2.8620	1.5324	-3.9118	8.0370	3.0432	1.4799	-3.1584	8.4206	3.2217	1.4391	-1.7551	8.5627
	Meso-regions	134	2.6716	1.7363	-2.2713	6.8523	2.9145	1.6534	-2.0703	7.2837	3.1393	1.5847	-1.6134	7.4888
	States	27	2.2483	1.9078	-1.7278	5.3361	2.5998	1.7710	-1.0676	5.5635	2.9022	1.6299	-0.0560	5.6895
	MCAs*	3657	7.4927	0.8118	3.0662	9.6175	7.1168	0.8686	2.3026	9.5129	7.0159	0.8657	2.3026	9.3276
In (transmortation cost to CD)	Micro-regions	522	7.6217	0.8696	3.8157	9.6053	7.2534	0.9409	3.8992	9.5005	7.1359	0.9244	3.7987	9.3134
ווו (נומוואטט ומנוטון בטאר נט או /	Meso-regions	134	7.7099	0.8401	5.1427	9.5446	7.3438	0.9389	4.7374	9.4325	7.2140	0.9143	4.6706	9.2272
	States	27	8.0163	0.7670	6.1295	9.2782	7.6797	0.8808	5.4841	9.1287	7.5316	0.8359	5.3315	8.7342
Note: Own elaboration from c	lata of IBGE. * Minimu	m Comparab	ile Areas (MCAs)	). ** Population	growth (n) is ad	justed for deprec	iation (d) and t	technological gr	owth (g), under	the usual assump	otion that d+g	equals 0.05.		

APPENDIX

Appendix A

Discussion Paper

Very Evaluating Multiple Spatial Dimensions of Economic Growth in Brazil using Spatial Panel Data Models (1970-2000)

### FIGURE A.1





4(







Note: In the map, the ranges were defined using natural breaks (Jenks) intervals. Estimated individual fixed effects come from the spatial Durbin models (SDM).

### Ipea – Institute for Applied Economic Research

### PUBLISHING DEPARTMENT

**Coordination** Cláudio Passos de Oliveira

#### Supervision

Everson da Silva Moura Reginaldo da Silva Domingos

### Typesetting

Bernar José Vieira Cristiano Ferreira de Araújo Daniella Silva Nogueira Danilo Leite de Macedo Tavares Diego André Souza Santos Jeovah Herculano Szervinsk Junior Leonardo Hideki Higa

### Cover design

Luís Cláudio Cardoso da Silva

### Graphic design

Renato Rodrigues Buenos

The manuscripts in languages other than Portuguese published herein have not been proofread.

### Ipea Bookstore

SBS – Quadra 1 – Bloco J – Ed. BNDES, Térreo 70076-900 – Brasília – DF Brazil Tel.: + 55 (61) 3315 5336 E-mail: livraria@ipea.gov.br

Composed in Adobe Garamond 11/13.2 (text) Frutiger 47 (headings, graphs and tables) Brasília – DF – Brazil

#### Ipea's mission

Enhance public policies that are essential to Brazilian development by producing and disseminating knowledge and by advising the state in its strategic decisions.





Institute for Applied Economic Research

Secretariat of Strategic Affairs

